Author’s response to reviews

Title: The effect of changing movement and posture using motion-sensor biofeedback, versus guidelines-based care, on the clinical outcomes of people with sub-acute or chronic low back pain - a multicentre, cluster-randomised, placebo-controlled, pilot trial

Authors:

Peter Kent (pkent@health.sdu.dk)
Robert Laird (rob.laird@optusnet.com.au)
Terry Haines (terrence.haines@monash.edu)

Version: 2
Date: 9 December 2014

Author’s response to reviews:

The pages of the manuscript now have line numbers.

An Availability of Supporting Data section has been added.